Author's response to reviews

Title: Efficacy and safety of Vinflunine for advanced or metastatic urothelial carcinoma in routine practice based on the French multi-centre CURVE study

Authors:

Jacques Medioni (jacquesmedioni@gmail.com)
Mario Di Palma (Mario.DIPALMA@gustaveroussy.fr)
Aline Guyot (Aline.GUILLOT@icloire.fr)
Dominique Spaeth (oncomed@wanadoo.fr)
Christine Theodore (c.theodore@hopital-foch.org)

Version: 2 Date: 19 March 2015

Author's response to reviews:

We have now revised the manuscript according to your recommendations.
All lines of the manuscript have been numbered.
The title page has been added.
Conflict of interest have been added.
Authors’ Contributions have been added.